Author's response to reviews

Title: The Global Evidence Mapping Initiative: Scoping research in broad topic areas

Authors:

Peter Bragge (peter.bragge@monash.edu)
Ornella Clavisi (ornella.clavisi@monash.edu)
Tari Turner (tari.turner@monash.edu)
Emma Tavender (emma.tavender@monash.edu)
Alex Collie (alex.collie@monash.edu)
Russell L Gruen (r.gruen@alfred.org.au)

Version: 12 Date: 17 June 2011

Author's response to reviews:

a minor revision was required in the acknowledgements section, all other sections of the manuscript are as previous